Author’s response to reviews

Title: Centella asiatica Juice effects on DNA damage, Apoptosis and Gene Expression in Hepatocellular carcinoma

Authors:

  Faridah Hussin (dka_as@yahoo.com)
  Sima Ataollahi Eshkoor (simaataolah@yahoo.com)
  Asmah Rahmat (satirann@yahoo.com)
  Fauziah Othman (sima_d1974@yahoo.com)
  Abdah Akim (s1_ataolah@yahoo.com)

Version: 5 Date: 29 June 2013

Author’s response to reviews: see over
Date: 28 June 2013

The Executive Managing Director
BMC Complementary and Alternative Medicine

Dear Sir/Madam,

Subject: Submission of Manuscript
Type of Manuscript: Research Article

I would like to submit the attached manuscript for possible evaluation.

Manuscript Title;

“Centella asiatica Juice effects on DNA damage, Apoptosis and Gene Expression in Hepatocellular carcinoma”

Name and address of corresponding author:

Asmah Rahmat
Faculty of Medicine and Health Sciences
Universiti Putra Malaysia
43400 UPM, Serdang, Selangor, Malaysia
Email: satirann@yahoo.com

I affirm that the manuscript has been prepared in accordance with the Instructions to Authors.

I hereby affirm that the content of this manuscript or a major portion thereof has not been published in a refereed journal, and it is not being submitted for publication elsewhere. This is the first comprehensive report about the protective effects of Centella asiatica Juice against hepatocellular carcinoma.

Thank you,

Yours sincerely,

Professor Dr Asmah Rahmat
Corresponding Author